Dear Editor-in-Chief:
Thank you very much for your valuable comments and suggestions for our article, and we will re-review our manuscript and revise everything you propose one by one to meet the journal's publication requirements.

Reviewer #1:

**Question 1: Is sclerosing gland a precursor to prostate adenocarcinoma?**

**Answer:** No, because prostate sclerosing adenopathy is a proliferative lesion with very complex histology. While the literature on the risk of prostate cancer following a diagnosis of sclerosing adenopathy is very limited, to our knowledge, the simultaneous dysplasia of epithelial and mesenchymal areas in sclerosing adenopathy appears to underlie the subsequently increased likelihood of prostate malignancy, but whether sclerosing adenopathy can be used as an independent feature to predict the risk of prostate cancer in men, there is currently insufficient evidence to prove this, and a large multi-institutional prospective study is needed to confirm.

**Question 2. Although most patients with sclerosing adenopathy have PSA within the reference range, do the authors recommend long-term follow-up for these patients?**

**Answer:** Long-term follow-up observation is recommended. In this study, although the PSA values of most patients with sclerosing adenopathy were within the reference range, patients should be informed that long-term follow-up observation should be performed. When it increases, and there are clinical symptoms such as frequent urination, urgency, dysuria, hematuria, dysuria, etc., you should seek medical attention in time to worry about the possibility of prostate cancer based on sclerosing adenopathy.

Reviewer #2:

Minor revisions to comments made by reviewers

4. **Question: Language Polish Requirements for Revised Manuscripts Submitted by Authors Who Are Non-native English Speakers**

**Answer:** We have hired a professional team to polish the language to make it meet the
needs of the magazine

5 Question: Abbreviations

Answer: Abbreviations are not covered in this article

6. Suggestions for comments from the editorial office

(1) Scientific editors:

The authors discuss whether sclerosing adenopathy may be a precursor to prostate adenocarcinoma in the Discussion section and add reference number 31. And the document "Care Checklist - 2016" was resubmitted with the correct version and patient signature.

(2) The editor-in-chief of the company:

The images are original, we have reprocessed and submitted the images, and the form has been resubmitted to meet the needs of the magazine.